Author’s response to reviews

Title: Cortical GABAergic neurons is more severely impaired by alkalosis than acidosis

Authors:

Shuyan Zhang (zhangsy70@yahoo.com)
Piyun Sun (spy89902003@yahoo.cn)

Version: 4 Date: 18 November 2013

Author’s response to reviews:

Responses to Reviewer Two:

Question #1: The manuscript has improved markedly through the various clarifications and elaborations the authors made in the text and satisfy all concerns from this reviewer.

Level of interest: An article of importance in its field.

Quality of written English: Acceptable.

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: I declare that I have no competing interests’ below.

Response #1: We sincerely appreciate Reviewer two’s comments above.